Author's response to reviews

Title: Assessing the accessibility of HIV care packages among tuberculosis patients in the North-West Region, Cameroon.

Authors:

Nwarbébé Barnabas Njozing (njozing.barnabas@epiph.umu.se)
Miguel San Sebastian (miguel.sansebastian@epiph.umu.se)
Pius Muffih Tih (piustih@cbchealthservices.org)
Anna-Karin Hurtig (anna-karin.hurtig@epiph.umu.se)

Version: 6 Date: 5 March 2010

Author's response to reviews: see over
POINT- BY- POINT RESPONSE TO EDITORIAL COMMENTS

Major revisions:

1. The full affiliations to the various authors have been included in the title page of the manuscript.

2. It has been stated that all authors read and approved the final version of the manuscript in the “authors’ contributions” section on page 18.

3. All the references have been correctly cited in numerical order in the text and also arranged numerically in the reference section on page 19.

Minor revisions:

1. Keywords have been removed from the abstract as requested.

2. All the vertical lines have been removed from the tables and properly formatted.

3. The figure has also been properly formatted.

4. The manuscript has also been checked for typographical errors and corrected.